Author's response to reviews

Title: Streptococcus acidominimus causing invasive disease in human, a case series

Authors:

chaoneng wu Dr. (wuchao1107@126.com)
buqing lian Dr. (liangbuqing@gmail.com)
yunguo gong Dr. (gongyunguo123@gamil.com)
lan zhang Mrs. (wu.chaoneng@zs-hospital.sh.cn)
yunzeng zou Dr. (zou.yunzeng@zs-hospital.sh.cn)
junbo ge Dr. (ge.junbo2@zs-hospital.sh.cn)

Version: 4 Date: 28 November 2013

Author's response to reviews:

Dear edirots,

According to your suggestions, I have made the formatting changes required for our manuscript, one is to include all the co-authors' email addresses in the Title page; the other is to include the Ethnicity of the patient in the Abstract (Han Chinese people) and Case presentation sections (page 3, before the ending of the first paragraph).

thanks.

Chaoneng Wu